Author's response to reviews

Title: Nefopam for the prevention of perioperative shivering: A Meta-analysis of randomized controlled trials

Authors:

Meng Lv Mr (gylvmeng@163.com)
Xuetao Wang Mr (taoxuewang@vip.com)
Wendong Qu Mr (wendongqu@163.com)
Mengjie Liu MS (liumengjie31@163.com)
Yuelan Wang Professor (wyldgf@163.com)

Version: 3 Date: 10 May 2015

Author's response to reviews:

Dear editorial board of BMC anesthesiology,

Please find the revised manuscript: Nefopam for the prevention of perioperative shivering: a Meta analysis of randomized controlled trials, by Meng Lv et al., to be submitted as a research article to the BMC anesthesiology for consideration of publication. All co-authors have seen and agreed with the contents of the manuscript and there is no financial interest to report. We certify that the submission is original work and is not under review at any other publication.

In Page 1 Line 18, “taoxuewang@vip.com” was replaced with “taoxuewang@sina.vip.com”.
In Page 2 Line 8, “the side effects of nefopam” was deleted.
In Page 2 Line 17, “weight mean difference(WMD)” was changed to “weighted mean difference(WMD)”.
In Page 2 Line 29, “was” was replaced with “is”.
In Page 2 Line 30, “without causing any serious side effects” was changed to “without influencing the extubation time”.
In Page 3 Line 24, “the side effect of nefopam” was changed to “the influence of nefopam on the extubation time”.
In Page 5 Line 17, “anesthesia” was changed to “anaesthesia” and “analysis” to “analyses”.
In Page 5 Line 31, “52 studies” was changed to “53 studies”.
In Page 6 Line 3, “, which” was instead of “that”.
In Page 6 Line 4, “1124 participants” was changed to “925 participants”.
In Page 6 Line 8 and Line 9, “anaesthesia” was instead of “anesthesia”.
In Page 6 Line 11, “a large variety of” was changed to “multiple”.
In Page 6 Line 23, “peri-operative” was replaced with “perioperative”.

Please find the detailed corrections listed above.
In Page6 Line24, “with 794 patients” was deleted.
In Page8 line6 and line11, “peri-operative” was changed to “perioperative”.
In Page8 Line18, “anaesthesia” was instead of “anesthesia”.
In Page8 Line23, “patient” was instead of “patients”.
In Page9 Line10, “has” was added to this line.
In Page9 Line30, “Finally, in this meta-analysis, reviewers were not blinded to study authors, which might increase the risk of bias. In the future study, investigators should be blinded to study authors, institutions, journal of publication, and results of study. ” was added to the last paragraph of the discussion.
In Page10 Line5, “besides” was changed to “however”. “besides” was added to this line.
In page18, we have made some changes of Table 2.

Sincerely yours,

Meng Lv, on behalf of the authors,

Corresponding author: Yuelan Wang at Department of Anesthesiology, Qianfoshan hospital, Shandong University, Jinan, Shandong, China.